Author's response to reviews

Title: Evidence-informed policy formulation and implementation: Comparative case study of two national policies for improving health and social care in Sweden

Authors:

Helena Strehlenert (helena.strehlenert@ki.se)
Linda Richter-Sundberg (linda.sundberg@ki.se)
Monica E Nyström (monica.nystrom@ki.se)
Henna Hasson (henna.hasson@ki.se)

Version: 6 Date: 26 November 2015

Author's response to reviews: see over
Dear Prof Nick Sevdalis,

Thank you for the constructive comments. We have shortened the manuscript and the text is now in line with the journal manuscript length requirement.

We have trimmed the text in the Background section and moved some more material (the Study setting section and Table 1) to additional files. A summary of the findings in Table 1 is now provided in the narrative of the manuscript. We hope that you will find these revisions satisfactory.

Kind regards,

Helena Strehlenert

Helena Strehlenert, PhD student
Medical Management Center (MMC)
Department of Learning, Informatics, Management and Ethics (LIME)
Karolinska Institutet
171 77 Stockholm, SWEDEN
Mobile: +46-708-655641
E-mail: helena.strehlenert@ki.se